Author's response to reviews

Title: Talk, Trust and Time: A Longitudinal Study Evaluating Knowledge Translation and Exchange Processes for Research on Violence Against Women

Authors:

C Nadine Wathen (nwathen@uwo.ca)
Shannon L. Sibbald (ssibbald@uwo.ca)
Susan M. Jack (jacksm@mcmaster.ca)
Harriet L. MacMillan (macmilnh@mcmaster.ca)

Version: 7 Date: 30 August 2011

Author's response to reviews:

Authors’ contributions: added text: “All authors read and approved the final manuscript.”

Tables: you have cited tables 1-6 in the manuscript text, but have only provided 4 tables. Please provide tables 5 and 6 or remove their citations from the text.

DONE: the references to Tables 5 and 6 were to the small group tables that provided those quote – the word “Tables” (three instances) has been removed from these quotes and just left as the small group number.

Tables: we are unable to display shading and colours within the tables. Can we therefore ask you to use another method of highlighting the shaded text e.g. by making it bold. – DONE (used bold/italics to differentiate from previous line)

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript - DONE

Tables: please remove the visible vertical lines from your tables. DONE

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

DONE

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

DONE (to main text, figure and two additional files)